Author’s response to reviews

Title: Inhibition of biofilm formation in Mycobacterium smegmatis by Parinari curatellifolia leaf extracts

Authors:

Benjamin Bhunu (bhunubenjamin@gmail.com)

Ruvimbo Mautsa (ruemautsa@gmail.com)

Simbarashe Sithole (mandlasithole@gmail.com)

Stanley Mukanganyama (smukanganyama@medic.uz.ac.zw)

Version: 4 Date: 18 May 2017

Author’s response to reviews:

Please include a full Title Page at the beginning of your manuscript. Full details on the information that we would ask you to include in this section, as well as details on how to format it, can be found in our submission guidelines (https://bmccomplementalternmed.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article). We would also ask you to please ensure that you include the email addresses of all manuscript authors on your Title Page.

This has now been done

- Please remove both of your supplementary files from your manuscript. While we thank you for providing us with these, we do not consider it necessary for them to be published as additional files alongside your manuscript.

We have removed the supplementary files